Author’s response to reviews

Title: Periodontal disease, chronic kidney disease and mortality: Results from the Third National Health and Nutrition Examination Survey

Authors:

Ana C Ricardo (aricar2@uic.edu)
Ambarish Athavale (dr.ambarish@yahoo.com)
Jinsong Chen (jinsongc@uic.edu)
Hemanth Hampole (hhampole@uic.edu)
Daniel Garside (garside@uic.edu)
Phillip Marucha (marucha@ohsu.edu)
James P Lash (jplash@uic.edu)

Version: 2 Date: 24 February 2015

Author’s response to reviews:

All editorial requests regarding Title Page, line numbering and Conclusions section have been addressed in the revised manuscript.